Author’s response to reviews

Title: Expression and potential role of SNF5 in Endometrial Carcinoma

Authors:

Shiying Sun (sunsy71@hotmail.com)
Yi Wu (yikoulian_4272007@163.com)
Kai Zeng (zengkaiicc@163.com)
Yue Zhao (yzhao30@cmu.edu.cn)

Version: 1 Date: 10 Jan 2019

Author’s response to reviews:

Dear Dr. Ellen Menkhorst,

Thank you very much for your kind consideration on our manuscript submitted to BMC Women’s Health (BMWH-D-18-00720). Once again, we would like to take this chance to thank you and all the reviewers for the constructive and valuable comments which have helped us to significantly improve the quality of our manuscript. According to your suggestions, we have revised our manuscript very carefully. The following is a point-by-point response to the editorial and Fiona Cousins’s points.

(1) In addition to these, please replace Figure 1F, it is out of focus.

Reply: We have replaced Figure 1F with a well-focused image in our revised manuscript.

(2) In figure legends 2 and 3 please include the statistical test used and indicate whether the data shows SEM or Median etc.

Reply: We have included the statistical test used and indicated the details of the data in figure legends 2 and 3 in our revised manuscript.

Sincerely yours,
Shiyiing Sun, PhD

Department of Gynecology and Obstetrics

The Second Affiliated Hospital of Anhui Medical University

Hefei 230601, P. R. China

Tel: 86-183-2618-8617

Email: sunsy71@hotmail.com